Author's response to reviews

Title: Low back pain in junior Australian Rules football: a cross-sectional survey of elite juniors, non-elite juniors and non-football controls

Authors:

Wayne Hoskins (waynehoskins@iinet.net.au)
Henry Pollard (hpollard@optushome.com.au)
Chris Daff (chirodaff_013@hotmail.com)
Andrew Odell (boubblytang@hotmail.com)
Peter Garbutt (enhancehealthcare@iinet.net.au)
Andrew McHardy (golfinjury@optusnet.com.au)
Kate Hardy (katie.hardy@optusnet.com.au)
George Dragasevic (dragasev@tpg.com.au)

Version: 3 Date: 12 July 2010

Author's response to reviews:

Dear the editor,

Please find attached a final revised manuscript which has the corrections suggested by the associate editor included. These include a final proof read for typo's and grammatical errors and improvements in the presentation of data (in text and tables).

Regards,

Wayne